Author's response to reviews

Title: Educational outreach to general practitioners reduces children's asthma symptoms: a cluster randomised controlled trial

Authors:

Merrick Zwarenstein (merrick.zwarenstein@ices.on.ca)
Angeni Bheekie (abheekie@uwc.ac.za)
Carl Lombard (carl.lombard@mrc.ac.za)
George Swingler (swingler@ich.uct.ac.za)
Rodney Ehrlich (ehrlich@cormack.uct.ac.za)
Martin Eccles (Martin.Eccles@newcastle.ac.uk)
Michael Sladden (m.sladden@doctors.org.uk)
Sandra Pather (sandra.pather@absamail.co.za)
Jeremy Grimshaw (jgrimshaw@ohri.ca)
Andrew D Oxman (oxman@online.no)

Version: 4 Date: 21 September 2007

Author's response to reviews:

Indipreaccept team,

The formatting changes you requested have been completed.

Thank you,

Merrick Zwarenstein